Author's response to reviews

Title: Treating cutaneous leishmaniasis patients in Kabul, Afghanistan: cost-effectiveness of an operational program in a complex emergency setting

Authors:

Richard Reithinger (rreithinger@yahoo.co.uk)
Paul G Coleman (paul.coleman@lshtm.ac.uk)

Version: 5 Date: 30 January 2007

Author's response to reviews: see over
To whom it may concern:

Please find enclosed the revised version of MS 1767261107111933 [Reithinger and Coleman. Treating cutaneous leishmaniasis patients in Kabul, Afghanistan: cost-effectiveness of an operational program in a complex emergency setting] formatted according to Biomed Central guidelines.

Should you have any further query, please do not hesitate to contact me.

On behalf of all authors, I look forward to hearing from you soon.

Best regards,

Richard Reithinger, PhD
Epidemiologist
Clinical Trials Area
SPECIFIC CHANGES MADE TO THE MANUSCRIPT

MS 1767261107111933 [Reithinger and Coleman. Treating cutaneous leishmaniasis patients in Kabul, Afghanistan: cost-effectiveness of an operational program in a complex emergency setting].

Reviewer 1 (Theo Vos, TV) made one comment:

Minor comment

1. TV noted that –in line with good practice in economic evaluation- results presented in the manuscript should be limited to two significant digits only.

This has been amended in the revised version of the manuscript in both abstract and discussion.